Author's response to reviews

Title: Validating the British Columbia Perinatal Data Registry: a chart re-abstraction study

Authors:

Gillian Frosst (gillian.frosst@inteiorhealth.ca)
Jennifer Hutcheon (jhutcheon@cfri.ca)
KS Joseph (ksjoseph@cfri.ca)
Brooke Kinniburgh (BKinniburgh@phsa.ca)
Cathe Johnson (cjohnson@phsa.ca)
Lily Lee (lily.lee@phsa.ca)

Version: 4  Date: 12 May 2015

Author's response to reviews:

Please see cover letter attached.